Author's response to reviews

Title: Spatial distribution and characteristics of injecting drug users (IDU) in five Northeastern states of India

Authors:

Gajendra Kumar Medhi (gkmedhi_ro@rediffmail.com)
Jagadish Mahanta (jmahanta@hotmail.com)
Rajatashuvra Adhikary (radhikary@fhiindia.org)
Brogen S Akoijam (brogen@rediffmail.com)
Buno Liegise (bunlieg@yahoo.com)
Kalpana Sarathy Desai (kalpanasarathy13@gmail.com)
Chelliah Joshua Thomas (cjoshuathomas@gmail.com)
Bhupen Sarmah (bsarmah@hotmail.com)

Version: 6 Date: 24 January 2011

Author's response to reviews:

I am submitting herewith the revised manuscript with the following changes to the format of the manuscript as requested:-

1) Title page has been added as per requirement
2) Authors’ names in the manuscript are exactly same as given in the submission system
3) Full affiliations of the authors have been given
4) Corresponding authors names have been indicated
5) Title of the manuscript has been written in sentence case
6) Vertical lines of the tables have been removed
7) Few minor typing errors have been corrected

Sincerely Your',
G K Medhi